CSCE 970: Pattern Recognition Spring Semester, 1999

Homework Assignment 0

January 11, 1999 Due Date: January 25

1. (10 pts) Send e-mail to sscott@cse with exactly the following in the subject

CSCE 970 hwk 0

and exactly the following in the body of the message:

970 SSN addr alias name survey

where name is your name, SSN is your student ID number, addr is your e-mail address (you may leave off the .unl.edu), survey is your response to the survey (described below) and alias is an alias that you will use to identify yourself when I post grades on the web page. (I use the alias in lieu of social security numbers.) If you do not wish to have your grades posted, make your alias “NO”. Also, do not place any whitespace within any of the fields.

The value of the survey field should be yes or no depending on whether or not you have exclusive (or nearly exclusive) access to a PC with at least the following specifications: 486 or faster CPU, ˘ 8 MB RAM, ˘ 5 MB free hard disk space, and Windows 95 or Windows NT 3.51. These are the minimum requirements for a software package I am considering using for the computing assignments of this course. Your responses will tell me how many of you can install this package and use it on your own machines. From this we can determine how many copies need to be installed on the CS department’s machines (no site licenses are available, only individual ones).

Please send me this e-mail ASAP so you will be placed on the class e-mail list and I will have the survey responses soon.

2. (10 pts) Have a picture taken of yourself, write your name on the back of it, and deposit it in the envelope outside my office (305 Ferguson). This is to help me learn your names. If there are multiple people in the photo, clearly indicate which one is you.